Name of journal: *World Journal of Clinical Cases*

**Manuscript NO:** 84307

**Title:** Traumatic pancreatic ductal injury treated by endoscopic stenting in a 9-year-old boy: A case report

**Provenance and peer review:** Unsolicited Manuscript; Externally peer reviewed

**Peer-review model:** Single blind

**Reviewer’s code:** 00058403

**Position:** Editorial Board

**Academic degree:** MD

**Professional title:** Associate Professor

**Reviewer’s Country/Territory:** Brazil

**Author’s Country/Territory:** South Korea

**Manuscript submission date:** 2023-03-07

**Reviewer chosen by:** AI Technique

**Reviewer accepted review:** 2023-03-12 00:41

**Reviewer performed review:** 2023-03-13 01:36

**Review time:** 1 Day

<table>
<thead>
<tr>
<th>Scientific quality</th>
<th>[ Y ] Grade A: Excellent</th>
<th>[ ] Grade B: Very good</th>
<th>[ ] Grade C: Good</th>
</tr>
</thead>
<tbody>
<tr>
<td></td>
<td>[ ] Grade D: Fair</td>
<td>[ ] Grade E: Do not publish</td>
<td></td>
</tr>
</tbody>
</table>

<table>
<thead>
<tr>
<th>Novelty of this manuscript</th>
<th>[ Y ] Grade A: Excellent</th>
<th>[ ] Grade B: Good</th>
<th>[ ] Grade C: Fair</th>
</tr>
</thead>
<tbody>
<tr>
<td></td>
<td>[ ] Grade D: No novelty</td>
<td></td>
<td></td>
</tr>
</tbody>
</table>

<table>
<thead>
<tr>
<th>Creativity or innovation of this manuscript</th>
<th>[ Y ] Grade A: Excellent</th>
<th>[ ] Grade B: Good</th>
<th>[ ] Grade C: Fair</th>
</tr>
</thead>
<tbody>
<tr>
<td></td>
<td></td>
<td></td>
<td></td>
</tr>
</tbody>
</table>
**SPECIFIC COMMENTS TO AUTHORS**

The case report is interesting and well described. One question, is it about the weight (BMI/age) of the boy? Could this variable be a limitation for the endoscopic procedure?

Thank you for your comment.

His height, body weight, and BMI were 125 cm, 23 kg, and 14.7, respectively.

I do not think that BMI is a criterion for determining whether an endoscopic procedure can be performed. I believe that the experience and skills of the endoscopist are rather more important.
Name of journal: World Journal of Clinical Cases
Manuscript NO: 84307
Title: Traumatic pancreatic ductal injury treated by endoscopic stenting in a 9-year-old boy: A case report
Provenance and peer review: Unsolicited Manuscript; Externally peer reviewed
Peer-review model: Single blind
Reviewer’s code: 05531699
Position: Peer Reviewer
Academic degree: MD
Professional title: Doctor
Reviewer’s Country/Territory: Italy
Author’s Country/Territory: South Korea
Manuscript submission date: 2023-03-07
Reviewer chosen by: Geng-Long Liu
Reviewer accepted review: 2023-03-14 09:20
Reviewer performed review: 2023-03-18 16:39
Review time: 4 Days and 7 Hours

<table>
<thead>
<tr>
<th>Scientific quality</th>
<th>[ ] Grade A: Excellent</th>
<th>[ ] Grade B: Very good</th>
<th>[Y] Grade C: Good</th>
</tr>
</thead>
<tbody>
<tr>
<td></td>
<td>[ ] Grade D: Fair</td>
<td>[ ] Grade E: Do not publish</td>
<td></td>
</tr>
</tbody>
</table>

<table>
<thead>
<tr>
<th>Novelty of this manuscript</th>
<th>[ ] Grade A: Excellent</th>
<th>[Y] Grade B: Good</th>
<th>[ ] Grade C: Fair</th>
</tr>
</thead>
<tbody>
<tr>
<td></td>
<td>[ ] Grade D: No novelty</td>
<td></td>
<td></td>
</tr>
</tbody>
</table>

<table>
<thead>
<tr>
<th>Creativity or innovation of this manuscript</th>
<th>[ ] Grade A: Excellent</th>
<th>[Y] Grade B: Good</th>
<th>[ ] Grade C: Fair</th>
</tr>
</thead>
<tbody>
<tr>
<td></td>
<td></td>
<td></td>
<td></td>
</tr>
</tbody>
</table>
**SPECIFIC COMMENTS TO AUTHORS**

This is a well-presented case with some educational value. Only the conclusions are a bit overstated, and should be toned down. It is quite hard to draw any firm conclusion from a single case.

Thank you for your comment.

We revised the conclusion according to your suggestion.

In summary, we believe that endoscopic stenting of pancreatic ductal injury might be a feasible technique in children with traumatic pancreatic duct injury to avoid unnecessary operation.

In summary, we believe that endoscopic stenting of pancreatic ductal injuries may be a feasible technique in certain cases of children with traumatic pancreatic duct injuries to avoid unnecessary operations.